Author's response to reviews

Title: T-SPOT.TB responses during treatment of pulmonary tuberculosis

Authors:

Samantha Ribeiro (sbrmicroimuno@yahoo.com.br)
Kelly Dooley (kdooley1@jhmi.edu)
Judith Hackman (hackmanj@jhmi.edu)
Carla Loredo (loredoc@hucff.ufrj.br)
Anne Efron (aefron@jhmi.edu)
Richard E Chaisson (rchaiss@jhmi.edu)
Marcus B Conde (marcusconde@hucff.ufrj.br)
Neio Boechat (n_boechat@yahoo.com)
Susan E Dorman (dsusan1@jhmi.edu)

Version: 4 Date: 30 January 2009

Author's response to reviews:

The references have been revised to now include that the parent study report is "in press" at The Lancet.